Author's response to reviews

Title: The FLASSH study: protocol for a randomised controlled trial evaluating falls prevention after stroke and two sub-studies.

Authors:

Frances A Batchelor (f.batchelor@nari.unimelb.edu.au)
Keith D Hill (keith.hill@nh.org.au)
Shylie F Mackintosh (shylie.mackintosh@unisa.edu.au)
Catherine M Said (cathy.said@austin.org.au)
Craig H Whitehead (craig.whitehead@rgh.sa.gov.au)

Version: 5 Date: 31 March 2009

Author's response to reviews:

Minor revision made to remove formatting in one line which had been applied due to "track changes".